Tohoku University Research Center for Policy Design Discussion Paper

TUPD-2025-010

What Makes the Oil Pricing Center? The Impact of Futures Markets and Production

Junlian Gong

Graduate School of Economics and Management, Tohoku University

Jun Nagayasu

Graduate School of Economics and Management, Tohoku University

Jun 2025

TUPD Discussion Papers can be downloaded from:

https://www2.econ.tohoku.ac.jp/~PDesign/dp.html

Discussion Papers are a series of manuscripts in their draft form and are circulated for discussion and comment purposes. Therefore, Discussion Papers cannot be reproduced or distributed without the written consent of the authors.

What Makes the Oil Pricing Center? The Impact of Futures Markets and Production

Junlian Gong^{*} Jun Nagayasu[†]

June 1, 2025

Abstract

This study examines the impact of oil futures markets and production on the connectivity and speed of information transmission in a country's spot oil market within the global network. First, we estimate the causal relationships between 12 spot oil markets using the existence and strength of transfer entropy as edges and weights to construct a series of dynamic networks for the global oil market. Second, we use a temporal network model to analyze changes in the connectivity, closeness, and betweenness of each oil market over different periods, and then compare these variations. Our findings indicate that Brent serves as the central hub of the global oil market, followed by the West Texas Intermediate and Minas markets. Moreover, the presence of oil futures markets significantly enhances the connectivity, information transmission speed, and hub role of spot markets. Oil production also positively impacts connectivity and betweenness; however, it does not have a significant relationship with the speed of information transmission.

JEL classification: F3, G1

Keywords: Oil price; Temporal centrality; Financial network; Pricing center

^{*}Tohoku University, Graduate School of Economics and Management; Email: gong.junlian.r1@dc.tohoku.ac.jp

⁺Tohoku University, Graduate School of Economics and Management, Address: 27-1 Kawauchi, Aoba-ku, Sendai, Miyagi 980-8576 JAPAN. Email: jun.nagayasu.d8@tohoku.ac.jp, Tel: +81 22 795 6265, Fax: +81 22 795 6270.

1 Introduction

Since the 1980s, the presence of the West Texas Intermediate (WTI) and Brent crude oil futures markets has allowed these benchmarks to dominate global oil pricing.¹ Therefore, researchers have traditionally focused on these two benchmark oil prices (Frino et al., 2018; Scheitrum et al., 2018; Elder et al., 2014; Kaufmann and Ullman, 2009). Although electronic trading and capital flows have promoted price convergence between regions, often referred to as the "law of one price" (Bachmeier and Griffin, 2006; Bentzen, 2007), significant differences remain in the influence and price-setting capabilities of various oil markets. Studies using risk spillover effects to measure intermarket relationships have consistently indicated that Brent and WTI exhibit the strongest spillover effects (Chatziantoniou et al., 2023; Cui and Maghyereh, 2023; Lee et al., 2023). Graph theory and complex network analysis have yielded similar conclusions, reinforcing the centrality of these two benchmarks in the global oil market (Ji and Fan, 2016).

The heterogeneity in the oil markets has become more pronounced in recent times. Since 2004, oil prices have begun to reflect demand-driven information from emerging market economies, signaling a significant shift in traditional market dynamics (Ji and Fan, 2015; Ji and Zhang, 2019). In particular, variations in crude oil characteristics—including density, sulfur content, and origin—have led to the emergence of numerous oil products and markets worldwide. For example, the UAE and China launched new oil markets in 2007 and 2018, respectively, establishing benchmark prices distinct from those of Brent and WTI. This trend highlights the evolving competitive landscape of the global oil market, where the importance of oil in industrial activities has driven nations to seek control over oil pricing. Moroever, extreme events such as the Organization of the Petroleum Exporting Countries (OPEC) "Price War" and the COVID-19 pandemic have significantly distorted the competitive structure and supply–demand balance of the oil market. As a result, despite the increasing integration of the global oil market, disparities remain in the influence and pricing power of different oil markets (Ji

¹WTI crude oil is a light, low-sulfur crude oil primarily sourced from the U.S. and widely traded on the New York Mercantile Exchange. The Brent market is based in the UK and extracts crude oil from the North Sea region.

and Fan, 2016; Cui and Maghyereh, 2023).

These events have heightened the urgency of exploring the fundamental factors underlying the establishment of influential and more stable oil markets, and energy policymakers and market participants have begun to strategically prioritize gaining pricing advantages and enhancing the influence of national oil markets. Achieving dominance in the increasingly competitive global oil market requires a keen understanding of the key factors that determine oil pricing power.

In this connection, two issues need to be addressed to fill the aforementioned research gap. First, an in-depth investigation of whether traditional benchmarks still dominate the global oil pricing system is missing. Second, few empirical studies have investigated the factors underlying the pricing power and centrality of traditional benchmarks in the global market. To address these two issues, we empirically examine the roles of oil futures markets and production in shaping the connectivity and information transmission capacity of global spot oil markets. Drawing on the theory of commodity pricing power within the framework of international political economy (Strange, 1988), we use transfer entropy to quantify the causal relationships and intermarket linkages between 12 spot oil markets, and construct dynamic, temporal network models thereof. Our analysis employs the measures of network centrality-namely, temporal degrees, closeness, and betweenness-to assess connectivity, information transmission speed, and the role of the hub within the global oil network. We use a panel model to distinguish between the impact of oil production and the financialization of oil markets on the pricing power of spot oil markets.

Our findings provide novel insights: the establishment of oil futures markets significantly enhances connectivity, information transmission speed, and hub roles, collectively reinforcing the centrality of spot markets within the global oil network. In contrast, while oil production positively affects market connectivity, it has less significant effects on information transmission speed than financialization. These findings contribute to the understanding of the mechanisms that drive the influence of the oil market; they provide important implications for policymakers and market participants seeking to strengthen their national oil pricing power and ensure energy security.

The remainder of this paper is organized as follows. Section 2 reviews the relevant literature, Section 3 outlines the underlying economic theory, and Section 4 describes the research methodology. Section 5 explains our data set, Section 6 presents the empirical results, and Section 7 concludes.

2 Literature Review

Given the significance of oil for consumers and producers, the study and prediction of oil price behavior is a major subject of research. Based on the vector autoregression framework, researchers have employed the Diebold-Yilmaz spillover Diebold and Yilmaz (2009) and time-varying parameter vector autoregression models to analyze the dominance of various markets from the perspective of risk spillovers. Cui and Maghyereh (2023) used paired risk spillovers from 12 spot oil prices to construct a static oil network for their entire sample and analyzed the influence of each market. They showed that the risk spillover capabilities of various oil markets vary over time, revealing the possibility of a change in the influence and benchmark status of the oil markets over time. Moreover, graph theory-based analyses can address complexities arising from the increasing number of nodes in the global oil market. In an innovative approach, Zhang et al. (2014) used the intensity of direct competition between oil markets as an edge weight to establish an oil trade network. They were the first to introduce the role of competition in oil network analysis, suggesting that the economic characteristics of oil markets, countries, and regions can be incorporated into cross-border influence studies. Their findings indicate that Brent and WTI are the most influential markets.

Moreover, Zhu et al. (2023) used panel data to study the impact of internal and external risk spillovers on energy security in 46 economies and found that crossborder spillover effects only induce risks through price channels. Their study is a particularly valuable attempt to analyze cross-border risk spillovers in energy prices from internal economic factors. Zhu et al. (2021) analyzed the impact of oil trade patterns on the effects of transnational systemic risk spillover in the global oil market, using Gaussian graphical models to estimate systemic risk spillovers between the oil sectors in 27 countries and describe oil trade patterns as a complex network. Results from their panel logit model showed that the intensity of bilateral trade and import competition could enhance oil market risk spillovers. Overall, risk events in North America and Europe trigger stronger spillovers than in Asia and vary over time.

Some studies have analyzed the adjustment speeds of multiple global market prices to illustrate the differences between markets. For example, Kaufmann (2016) used an error correction model (ECM) to analyze the adjustment speeds of 33 spot oil prices when deviating from the long-term equilibrium. The deviation of a market's price from the long-term equilibrium owing to national risk was defined as the cost of choosing that market. Accordingly, oil prices from countries with higher national risks adjusted more slowly, and factors such as sulfur content, density, distance between supply ports, and OPEC membership affected the ability to adjust the price of oil markets. Kaufmann's study presents information on the factors that influence the ability to adjust cross-border prices in the oil markets.

Both the financialization of the oil markets and oil production can explain heterogeneity in the oil markets. Financialization is particularly important in the analysis of oil price fluctuations; it reflects the impact of capital flows on oil asset prices, speculation and hedging, and the derivative price discovery function (Feng et al., 2022). The WTI oil market is more financially integrated compared with the Brent oil market, which is mainly influenced by global changes in supply and demand and less influenced by financialization and speculation (Tudor and Anghel, 2021). Hirshleifer (1988, 1989, 1990) focused on the price discovery function and effects of the information spillover of derivative markets, represented by oil futures.

Futures markets are known to be closely linked to spot markets. Silverio and Szklo (2012) used the Kalman filter technique to obtain temporal metrics of the contribution of the WTI oil futures market to the spot price discovery mecha-

nism. Kaufmann and Ullman (2009) demonstrated that the futures prices of WTI and the Dubai crude oil spot prices are Granger-causative factors for a series of oil prices, and contribute to the rigorous definition of oil price influence. Compared with risk spillovers, the existence and strength of causal relationships better explain the influence of oil markets. However, Granger causality tests are not effective in describing the specific strength of causal relationships between time series. Transfer entropy can integrate the advantages of describing influence magnitude, as in net risk spillovers, and explaining causal relationships, as in Granger causality tests. This method has been widely used to study information transmission between financial markets. Zhou et al. (2024) used multiscale entropy methods to analyze interactions between innovative and traditional financial assets on short-, medium-, and long-term scales. Niu and Hu (2021) used transfer entropy networks to measure the transmission of information and the relationships between the Chinese stock market and the commodity futures market on different time scales. They found that the direction and magnitude of information spillovers vary with asset types and time scales. These studies emphasize the superiority of entropy methods in explaining information transmission.

In addition to financialization, the advantages of production are equally important in explaining market heterogeneity. Razek and Michieka (2019) used the vector autoregression model to study the impact of production by OPEC and non-OPEC countries, global and Chinese demand, and oil financial assets on oil prices. They reiterated the importance of production levels in the U.S. and OPEC. De Graaff (2012) found that the expansion and production cooperation of stateowned oil companies from countries such as Russia promoted the multipolarity of the global oil order. Wu et al. (2022) also emphasized the role of production advantages in the multilayered network risk transmission mechanisms of international energy enterprises.

In this study, we employ a dynamic network approach based on transfer entropy, which extends the traditional pairwise causal analysis of transfer entropy to a high-dimensional, multi-node setting. This methodological advancement allows for a more rigorous and nuanced definition of the "pricing center" compared with existing studies. The varying roles of market nodes can also be distinguished using network metrics such as temporal degree, temporal closeness, and temporal betweenness in order to capture the diverse behaviors observed in market dynamics. Importantly, our dynamic network analysis is grounded in graph theory. Unlike the quantile-based risk spillover network in Zhou et al. (2022), our method uncovers the underlying market structure rather than merely assessing the influence of individual markets. Ji and Fan (2016) also employed graph theory, using a minimum spanning tree to analyze connectivity and global market structure. Our study goes beyond simple connectivity by incorporating a broader range of network indicators and examining their respective driving factors. Overall, we comprehensively investigate how financial futures markets and production-related factors influence the role of oil markets within the global oil network.

3 Economic Theory

The price information can flow invisibly and rapidly through electronic systems in different markets worldwide. A country with an efficient oil pricing center can access detailed information about oil price changes in real time, gaining an information advantage over other countries or markets in controlling oil price performance. From the perspective of financial economics, Fama (1970), in the "Efficient Market Hypothesis," proposed that efficient markets quickly reflect all available information, thereby reducing asset pricing errors and optimizing capital allocation. In this regard, all investors hope to trade in a fully efficient market to achieve their investment goals. We use the physical and statistical concept of transfer entropy to measure the directed flow of information between these different oil market nodes (Schreiber, 2000). Specifically, the advantage of statistical information prediction between two price series can be seen as the strength of the transfer entropy. When one market consistently exhibits transfer entropy with respect to another, we can view it as information flowing from that market, and that such a market influences other markets.

We accordingly extend the scenario from two markets to multiple markets, where dynamic networks perfectly meet our needs. In the physical properties of dynamic networks, "temporal degree" represents the average number of information flows from one market to all other nodes in the network, reflecting the connectivity of markets (Kaufmann et al., 2004). The strength of the information flow represented by transfer entropy is used to measure the distance between the nodes in the network. Then, the inverse of the average distance from each market node to the other nodes becomes the "temporal closeness." From a physical perspective, the market node with the highest temporal closeness is the global oil pricing center, which can exchange and transmit information the fastest. We view this as the most efficient market described by Fama (1970). Another advantage of the network model is that it captures the intermediary role of markets through "temporal betweenness," which reflects how often a market lies on the shortest information paths between others. Markets with high betweenness act as key bridges in the global oil network, facilitating information flow between otherwise disconnected regions.

We also examine the factors that cause differences in the information transmission and pricing ability of different markets, with financial markets represented by derivatives and production being the most concerning. On the one hand, Strange (1988) summarized the factors that influence a country's global commodity pricing power from the perspective of political economy: financialization, production, security, and knowledge. Strange (1988) argued that financialization enhances the market's price discovery function and stability by providing liquidity, risk management tools, and capital flows. Her point of view is highly macroeconomic, while we focus on the oil futures market, which is an aspect of financialization as described by Strange (1988). On the other hand, production structures directly influence prices by controlling supply, production costs, and commodity quality. The combination of these two factors determines a country's or entity's pricing power in the global market and shapes the dynamics and influence of commodity prices.

Inspired by Scheitrum et al. (2018), we integrate the widely recognized view

in financial economics that futures markets contribute to the discovery of spot prices; we focus on the establishment of the financial futures market, a key aspect of financialization. Garbade and Silber (1983) provided a quantitative method to evaluate the contribution of the information flow from the futures market to the spot price discovery process. Schwarz and Szakmary (1994) combined the statistical methods of Garbade and Silber (1983) and Engle and Granger (1987) to incorporate the influence of the futures market on the spot market in cointegration tests.

These studies used cointegration and the error correction model to examine the relationship between crude oil spot prices and futures prices, and defined the temporal contribution of futures prices to the spot price discovery process.

Following the literature, the contribution of the futures market to the spot market's price discovery process can be measured by representing the relationship between the two variables as a cointegration model with error correction. Assume that S_t and F_t are the logarithms of the spot and futures prices of the commodity at time t, respectively, and β_t synthesizes the elements proposed by the theories discussed earlier. The cointegration relation is expressed as

$$S_t = F_t + \beta_t \tag{1}$$

The error correction model can be written as

$$\Delta F_{t} = \alpha_{F} \left(S_{t-1} - F_{t-1} - \beta_{t-1} \right) + c_{F1} \Delta S_{t-1} + c_{F2} \Delta F_{t-1} + \epsilon_{F_{t}}$$
(2)

$$\Delta S_{t} = \alpha_{S} \left(S_{t-1} - F_{t-1} - \beta_{t-1} \right) + c_{S1} \Delta S_{t-1} + c_{S2} \Delta F_{t-1} + \epsilon_{S_{t}}$$
(3)

In these equations, S_t and F_t are the logarithms of the WTI spot and firstmonth futures prices, respectively. α_S and α_F are the adjustment coefficients in the error-correction equations for the spot and futures markets, respectively. c_{F1} , c_{F2} , c_{S1} , and c_{S2} are the coefficients of deterministic terms. ϵ_{F_t} and ϵ_{S_t} are uncorrelated white noise residuals. Caporale et al. (2014) explained the contribution of the futures price to spot market price discovery as δ , which can be calculated as:

$$\delta = \frac{\alpha_S}{\alpha_S + \alpha_F} \tag{4}$$

However, as δ is a point estimate derived from the error correction model, it does not capture the time-varying nature of the market's contribution to the price discovery process. To address this, Foster (1996) and Caporale et al. (2014) proposed using a state–space model combined with the Kalman filter technique to analyze fluctuations in the adjustment coefficients, thereby obtaining a timevarying measure of price discovery (δ_t). The state–space model's transition equations, which describe the evolution of the system's state over time, are as follows:

$$A_{1t} = A_{1t-1} + \eta_{1t} \tag{5}$$

$$A_{2t} = A_{2t-1} + \eta_{2t} \tag{6}$$

By applying the Kalman filter, which is a recursive algorithm, we can dynamically estimate the state variables A_{1t} and A_{2t} over time. The time-varying price discovery metric δ_t can then be calculated through the following steps:

1. Initialization: Set the initial values for the state variables $A1_0$ and $A2_0$ and their covariance matrix.

2. Prediction: Predict the next state variables using the state equation.

$$\begin{pmatrix} A1_{t|t-1} \\ A2_{t|t-1} \end{pmatrix} = \begin{pmatrix} A1_{t-1} \\ A2_{t-1} \end{pmatrix} + \begin{pmatrix} \eta 1_t \\ \eta 2_t \end{pmatrix}$$
(7)

3. Update: Update the state variables and covariance matrix using the observation data.

$$\begin{pmatrix} A1_t \\ A2_t \end{pmatrix} = \begin{pmatrix} A1_{t|t-1} \\ A2_{t|t-1} \end{pmatrix} + K_t \begin{pmatrix} \Delta F_t - \Delta F_{t|t-1} \\ \Delta S_t - \Delta S_{t|t-1} \end{pmatrix}$$
(8)

where K_t is the Kalman gain, balancing the error between prediction and observation.

4. Compute δ_t : Calculate δ_t using the updated state variables $A1_t$ and $A2_t$:

$$\delta_t = \frac{A2_t}{A2_t + A1_t} \tag{9}$$

By following these steps, the Kalman filter algorithm dynamically assesses the contribution of the futures market to the spot price discovery δ_t at each time point, capturing its time-varying nature.

A higher value of δ indicates that the futures market plays a greater role in price discovery, with futures prices leading spot prices. A lower value of δ suggests that the spot market dominates price discovery. By tracking the timevarying nature of δ , we can dynamically assess the contributions of the futures market. We regard this price discovery function as the flow of information from the futures market to the spot market. Therefore, spot markets with an associated futures market can access more information, allowing faster adjustments. In contrast, spot markets without futures markets are unable to obtain high-frequency supply and demand information, making it difficult to fully reflect the effective information of the global oil market; thus, they are less frequently referenced by investors. The market efficiency driven by this price discovery function is reflected in the dynamic network as a faster information exchange, that is, higher temporal closeness. This also enhances the information connectivity and hub function of the spot market. Therefore, we derive the following hypotheses:

Hypothesis 1: The oil futures market could enhance the information connectivity and transmission speed of spot market prices.

Production is the second factor examined in this study. Within the supplydemand framework, the share of production is the most important factor on the supply side. For example, OPEC holds a major share of global oil production, significantly monopolizing oil production and controlling oil prices. Kilian (2009) used a structural vector autoregression model to define the impact of supply, represented by production, on oil prices and confirmed the influence of production on the oil market. In practice, changes in production are the main regulatory tools that OPEC countries use to influence oil prices. Moreover, production predicts oil prices and alters market behavior. Monge et al. (2017) studied price behaviors in the WTI oil market before and after the U.S. shale oil revolution, confirming that the increase in production owing to this revolution led to a decrease in WTI oil prices in different frequency domains. When the production volume of a country or region increases, its share in the global oil market also increases. This means that the market engages in more transactions with other markets through the export of more oil, thereby enhancing the connections and interactions between markets. While increased production may improve the physical connectivity of the market, the speed of information flow is also influenced by market mechanisms, trading frequency, and information transparency.

Hypothesis 2: An increase in oil production can enhance the information spillover capacity of the oil market.

Thus, we identify two factors, the presence of the futures market and crude oil production, influencing information flows in international markets.

4 Methodology

4.1 Transfer Entropy

Transfer entropy is a nonparametric measure introduced by Schreiber (2000) to quantify the direction and amount of information transfer between two time series. It thus captures causal influence and nonlinear dependencies. For two time series *X* and *Y*, the transfer entropy from *X* to *Y* ($T_{X \to Y}$) is defined as:

$$T_{X \to Y} = \sum_{y_{t+1}, y_t, x_t} p(y_{t+1}, y_t, x_t) \log \frac{p(y_{t+1} \mid y_t, x_t)}{p(y_{t+1} \mid y_t)}$$

where y_t is the state of Y at time t. x_t is the state of X at time t; $p(y_{t+1}, y_t, x_t)$ is the joint probability distribution of Y at t + 1, Y at t, and X at t. $p(y_{t+1} | y_t, x_t)$ is the conditional probability of y_{t+1} given y_t and x_t ; $p(y_{t+1} | y_t)$ is the conditional probability of y_{t+1} given y_t .

Transfer entropy measures how much knowing the past states of X reduces the uncertainty in predicting the future states of Y, beyond what is already known from the past states of Y. In other words, it reveals the role of X as an information source for *Y*. If $T_{X \to Y} > 0$, it indicates that *X* provides additional information about *Y*, showing a directional influence from *X* to *Y*. If $T_{X \to Y} = 0$, it indicates that *X* does not provide any extra information for *Y*.

We use the strength of the transfer entropy as a key indicator to assess pricing power. The greater the transfer entropy, the stronger the influence of *X* as an information source for *Y*, indicating a higher level of inter-market influence.

4.2 Temporal Complex Network

We employ the complex network temporal centrality indicators developed by Kim and Anderson (2012) to examine the structure of the global crude oil market and changes in key nodes. Each crude oil market is considered as a node in the complex network, with the most representative crude oil product price serving as the data flow. Initially, we define the dynamic network with time parameters as $G_{(0,T)}^D = (V, E_{(0,T)})$ on a time interval [0, T], consisting of a set of vertices (V)and a set of temporal edges $E_{(0,T)}$, where a temporal edge $(u, v)_{(i,j)} \in E_{(0,T)}$ exists between vertices u and (V) on a time interval [i, j] such that $i \leq T$ and $j \geq 0$. In the dynamic network, (V) remains constant, while the set of existing edges may change over time.

When the time information is divided into sequences with a duration of w = T/n, we obtain a dynamic network consisting of a series of static graphs. Specifically, the notation G_t $(1 \le t \le n)$ represents the aggregate graph, which includes a set of vertices V and a set of edges E_t , where an edge $(u, v) \in E_t$ exists only if a temporal edge $(u, v)_{(i,j)} \in E_{(0,T)}$ exists between vertices u and v on a time interval [i, j] such that $i \le wt$ and j > w(t - 1). In other words, G_t is the t^{th} temporal snapshot of the dynamic network $G_{(0,T)}^D$ during the t^{th} time window. In each short snapshot, the directed graph G = (V, E) is set with vertices (v_t) for each $t \in \{0, 1, ..., \infty, n\}$, having edges from $u_{(t-1)}$ to v_t and vice versa for an edge $(u, v) \in E[t]$; and it has edges from $u_{(t-1)}$ to v_t for all $v \in V$ and for all $t \in \{1, ..., n\}$.

Temporal Degree

In a network constructed on the basis of transfer entropy, the degree repre-

sents how often a node acts as an information source for other nodes, corresponding to the number of edges directly connected to that node in the graph. In a dynamic network, a higher degree indicates more connections, reflecting the activity and influence of the node within the system.

Since network structures evolve over time, the degree of each node also changes dynamically. The indicator $D_{(i,j)}(v)$ captures the connectivity of a node in a specific time interval, thereby providing information on the temporal evolution of the network.

The formula for temporal degree is defined as follows:

$$D_{(i,j)}(v) = \frac{\sum_{t=i}^{j} 2D_t(v)}{2(|V|-1)m}$$
(10)

where $D_{(i,j)}(v)$ denotes the temporal degree of node v within the time interval [i, j]. $D_t(v)$ represents the degree of node v at time point t, indicating the number of edges directly connected to node v at time t. m = j - i is the length of the time interval. |V| denotes the size of the node set.

Temporal degree directly reflects the concept of "connectivity" because a higher temporal degree indicates that a market engages in more frequent and active information exchange with other markets. This suggests that the market is more structurally integrated into the global oil market network and plays a more significant role in the dissemination of price signals. Therefore, temporal degree serves as a key metric to evaluate how a market is connected to and influential over others within the structure of global oil price transmission.

Temporal Closeness

Temporal closeness measures a node's efficiency in disseminating information within a network. Nodes with higher closeness can reach other nodes more quickly, thereby exerting greater influence. Unlike temporal degree, temporal closeness considers not only direct connections but also the inverse of the shortest paths to all other nodes. Nodes with high temporal closeness—often referred to as "centers"—play a critical role in controlling and distributing information throughout the network.

The formula for temporal closeness centrality is defined as

$$C_{(i,j)}(v)' = \sum_{i \le t < j} \sum_{u \in V \setminus v} \frac{1}{\Delta_{(t,j)}(v,u)}$$
(11)

where $\Delta_{(t,j)}(v, u)$ represents the temporal shortest-path distance from node v to node u over the time interval [t, j]. If no temporal path exists from node v to node u during the time interval [t, j], $\Delta_{(t,j)}(v, u)$ is defined as ∞ . Importantly, $\Delta_{(t,j)}(v, u)$ differs from $\Delta_{(t,j)}(v, u)$, as the time-ordered graph G is a directed graph.

To facilitate the comparison of closeness values across different nodes, we can normalize the values within a standard range by dividing $C_{(i,j)}(v)$ by (|V| - 1)m. This yields the standardized temporal closeness:

$$C_{(i,j)}(v) = \frac{1}{(|V|-1)m} \sum_{i \le t < j} \sum_{u \in V \setminus v} \frac{1}{\sigma_{(t,j)}(v,u)}$$
(12)

where $\sigma_{(t,j)}(v, u)$ represents the temporal shortest-path distance from node v to node u, |V| denotes the size of the node set, and m = j - i is the length of the time interval.

This indicator corresponds to the concept of "information transmission speed" because a higher temporal closeness indicates that a market can reach and influence other markets more quickly through shorter information paths. Such markets can respond more rapidly to global oil price signals and can reflect and transmit price changes in a timely manner. Therefore, temporal closeness serves as a proxy for measuring the speed and efficiency of information transmission within the global oil network.

Temporal Betweenness

Temporal betweenness measures the extent to which a node acts as an intermediary in the network over different time periods—that is, the role it plays as a bridge in the information transmission paths connecting other nodes. It is often used to assess the structural importance of a node within the entire network (Ji and Fan, 2016). The formula is as follows:

$$B_{(i,j)}(v) = \sum_{i \le t < j} \sum_{\substack{s \ne v \ne d \\ s, d \in V}} \frac{\sigma_{(t,j)}(s, d, v)}{\sigma_{(t,j)}(s, d)}$$
(13)

where $B_{(i,j)}(v)$ represents the temporal betweenness of node v within the time interval [i, j]. Here, $\sigma_{(t,j)}(s, d, v) = |S_{(t,j)}(s, d, v)|$ denotes the number of temporal shortest paths from node s to node d passing through node v during the time interval [t, j]. $\sigma_{(t,j)}(s, d)$ represents the number of temporal shortest paths from node s to node d during the time interval [t, j].

Temporal betweenness reflects the concept of a "hub role" because markets with high temporal betweenness frequently lie on the shortest informational paths between other markets, effectively serving as intermediaries or bridges. Such markets play a critical role in connecting otherwise disconnected regions and facilitating the transmission of global oil price signals. Their position within the network grants them structural importance, enabling them to influence the flow and distribution of information across the entire system.

5 Oil Prices

We select 12 crude oil spot prices, covering Brent, WTI, and Russian crude oil, representing Europe and North America, as well as prices from the main oilproducing countries of the Middle East and East Asia. The sample period spans from January 29, 2010, to February 26, 2024, with a total of 734 weekly settlement prices included after removing one outlier. To analyze medium- and long-term changes in the global oil market, the chosen sample period encompasses several critical periods that have had a profound impact on oil prices. For instance, the "Price War" within OPEC in 2014 led to a surge in production and a sharp decline in oil prices. Although the OPEC strategy did not fully achieve its intended objectives, it contributed to the reallocation of global market shares and structural adjustments (Quint and Venditti, 2023; Ma et al., 2021). Another significant event was the COVID-19 pandemic, which led to a sharp contraction in global economic activity and a significant drop in oil demand in 2020, further driving profound changes in the supply–demand dynamics of the oil market. Finally, the Russia–Ukraine war disrupted the security of oil transport between Eastern and Western Europe, with the geopolitical risks arising from the war causing significant pressure on the Brent market, while the WTI market experienced relatively smaller fluctuations. This phenomenon indicates that regional geopolitical risks have, to some extent, distorted the competitive landscape of the global oil market (Almutairi et al., 2025).

We also include the Eastern Siberia–Pacific Ocean (ESPO) crude oil price data, a relatively under-analyzed Russian crude oil primarily exported to China and India. After the Russia–Ukraine war, the exports of Ural blend oil to Europe were significantly reduced; however, Russia increased its oil trade with China, leading to a rise in ESPO crude oil trade volumes. Analyzing the role of ESPO crude oil in the global network is not only novel, but it also provides important policy insights. Figure 1 illustrates the trends of the 12 crude oil spot prices, with extreme price declines observed in the global oil market from 2014 to 2016 and early 2020. Table 1 provides the background information sourced from DataStream.

We process all price series into logarithmic return series (Equation 14) to meet the assumptions and requirements of the statistical analysis. Figure 2 and Table 2 present the trends of the logarithmic return series and the descriptive statistical results, respectively. The kurtosis results are adjusted by subtracting the standard normal distribution kurtosis value of 3, and all are negative. This indicates a higher probability of extreme decreases in oil prices, with more extreme values on the negative side of returns. All return series significantly deviate from the extreme characteristics of the standard normal distribution, which is consistent with the findings of other studies on oil prices and returns (Fan et al., 2008; Zhang et al., 2019). All return samples pass the augmented Dickey—Fuller test at the 1% level, indicating that they can be considered stationary series.

$$R_t = \ln(P_t) - \ln(P_{t-1})$$
(14)

The representation for the return of the oil price at day *t* is the oil price at day *t*, and P_{t-1} is the oil price at day t - 1.

6 Empirical Results

6.1 Full Samples

We will now explore the centrality of a temporal network based on the returns of 12 spot prices for crude oil. First, we apply a 52-week (one-year) window to calculate the pairwise transfer entropy between the returns of 12 prices, with the selection of window size following the study on rolling transfer entropy networks by Choi and Kim (2024).² Since the transfer entropy is directional, each window generated 132 sets of transfer entropy results. We evaluate the robustness of the transfer entropy using the bootstrap method (300 times) and retain significant transfer entropy values at the 5% level as directed edges between two markets (Behrendt et al., 2019), with the transfer entropy value serving as the edge weight. This step completes the construction of the static network for the first window period.

Second, we shift the window forward by one week and recalculate the transfer entropy to construct a new network. Using this rolling-window method, we generate 682 networks over the sample period, which consist of 733 return data points. The network structure for each window period is represented by an adjacency matrix, which captures the connections between nodes within a specific time interval. Figure 3 illustrates the specific structures of the 1st, 10th, 100th, and 682nd network slices.

Based on the dynamic network composed of 682 time slices, we calculate the temporal degree, temporal closeness, and betweenness of each node, representing the number of connections, speed of information transmission, and mediating properties within each time window, respectively. To highlight the relative strengths of the network metrics for each node, we apply the min–max normalization to these indices, as shown in Equation (15):

²Choi and Kim (2024) employed 20, 60, and 120 days as window lengths to compute transfer entropy, constructing a series of dynamic network slices and verifying the existence of dynamic causal relationships between stock prices and trading volumes. Similarly, we select 52, 56, and 60 weeks as window lengths to demonstrate the robustness of the centrality measures of Brent and WTI under different parameter settings.

$$X' = \frac{X - X_{\min}}{X_{\max} - X_{\min}}$$
(15)

where *X* represents the original data value, *X'* the normalized data value, X_{\min} the minimum value in the dataset, and X_{\max} the maximum value in the data set. Normalization converts the data to a common scale, eliminating the dimensional differences between variables, facilitating the comparison of the relative advantages of different market pricing powers at different times. Unlike discrete rankings, min–max normalized data remain continuous while preserving the underlying data distribution, thereby enhancing the precision of the subsequent panel analysis.

Table 3 presents the normalized temporal degree, closeness, and betweenness results for each oil market across the entire sample under different window lengths. First, the connectivity of the Brent, Mexico, and WTI markets consistently ranks among the top four across all three window lengths. That is, these markets exhibit strong connectivity and innovation capabilities, enabling them to effectively exchange information with other markets. Second, as the window length extends from 52 weeks (one year) to 56 weeks (one year and one month), and then to 60 weeks (one year and two months), the Brent market maintains a leading position in closeness, followed by WTI, Minas, and Mexico. This suggests that these markets have the shortest average path length and the highest efficiency of information transmission within the network. In particular, the Mexico market, located in the same production region as WTI, demonstrates faster information transmission between the two markets. Hence, Brent and WTI are the most efficient markets in spreading price innovation information. Third, Brent consistently ranks first in terms of betweenness across all window lengths. Nodes with high betweenness typically act as intermediaries between different clusters, facilitating information flows that exhibit clear directionality and diversity. As a critical intermediary node, the Brent market plays a vital role in integrating and enhancing multidirectional information interactions. These findings underscore the hub-and-spoke roles of Brent and WTI in the global oil market network.

In addition to these prominent nodes, the temporal degree and closeness of

the Russian and Minas crude oil spot markets also rank highly. Russia's ESPO oil has become a key export product for the country, primarily transported to China. As a benchmark price in the Far East, Minas crude oil is low in sulfur, easy to process, and predominantly exported to Asian countries such as China and India. Finally, we observe that the information transmission capabilities of individual OPEC member countries are relatively low. Despite leading in total production, OPEC member countries have not yet established an international oil futures market. The total trading volumes of the Oman and Dubai oil futures markets are not comparable to those of Brent and WTI. In practice, OPEC and oil-producing countries such as Russia mainly rely on adjusting production levels to influence oil prices and maximize their national interests. The adjustment capabilities of individual countries are further constrained by geopolitical uncertainties in the Middle East and the operational constraints of OPEC organizations.

6.2 Rolling Samples

Using a rolling-window approach, we present the temporal centrality of various crude oil spot markets across different periods. Similarly to the full-sample analysis, we first calculate the results for each oil market in January 2012, covering the period from February 5, 2010, to January 27, 2012, for a total of 104 weeks (two years).³ During this sample period, we use a 52-week (one-year) window to calculate transfer entropy and establish 53 networks.

Then, we move the sample window according to the number of weeks in each month, resulting in 146 months of data collected from January 2012 to February 2024. Figures 4, 5, and 6 show the normalized temporal degree, closeness, and betweenness of each crude oil spot market over different periods, respectively. We also calculate the average values of each indicator over the entire sample period and find that the Brent market ranks first across all three metrics, followed closely by the WTI market, whereas some OPEC countries' crude oil spot markets

³Ji and Fan (2016) employed the graph theory approach of minimum spanning trees to analyze the evolution of the global oil market network; they used a rolling window of 100 weeks with a step size of 1. Following their methodology, we select a rolling window of 104 weeks (2 years) with a step size of 1 to construct a series of networks.

lag behind those of Brent and WTI. This is generally consistent with the findings for the full sample period.

The performance of each market varies across different periods. For example, between 2016 and 2018, the temporal degree, closeness, and betweenness of the Brent market declined, while the WTI, Minas, and Russian oil markets gained prominence. Specifically, the global oil market experienced a shift from oversupply to a supply–demand balance during this period. The OPEC+ production cut agreements, the resurgence of U.S. shale oil, and geopolitical uncertainties collectively contributed to the recovery of oil prices. However, the formation of OPEC+ did not stabilize the global oil market after 2017, as researchers observed changes in OPEC's behavior (Behar and Ritz, 2017). By early 2024, the centrality of different oil markets had undergone significant changes once again. The Brent market, particularly in Europe, faced geopolitical threats from the Russia–Ukraine conflict, while capital outflows from Europe further increased economic uncertainty. Consequently, the information sources and transmission mechanisms of the global oil market have experienced a structural shift (Zhang et al., 2024). During this phase, Malaysia's low-sulfur Tapis crude oil market briefly emerged as an important source of oil market information.

6.3 Impact of Futures Markets and Oil Production

Although we have assessed the spillover effects of information and pricing power in different markets, empirical discussions remain limited with respect to the economic factors that drive heterogeneity in the connectivity and information transmission capacity of crude oil spot markets. Therefore, before further analysis, we examine the relationship between the presence of futures markets and the concept of financialization.

Initially, we examine the impact of financialization from the perspective of financial derivatives. Our theoretical rationale is that futures markets perform a price discovery function for their corresponding spot markets and, therefore, have the potential to facilitate spot price adjustments. Specifically, if a crude oil spot market is associated with a corresponding futures market, we assign the value of 1; otherwise, it is set to 0. The resulting futures variable is also closely linked to a country's overall level of financial development.

To verify whether the presence of futures markets can serve as an appropriate proxy for financialization, we collected the Financial Development Index (FD) and the Financial Institutions Index (FI) for 12 countries from the IMF covering the period 1980–2021, and calculated their correlations with the futures market variable for each country. The FD index reflects countries' relative rankings in terms of the development, access, and efficiency of their financial institutions and financial markets. The FI index aggregates indicators from sectors such as banking, fund assets, and insurance and is widely used to measure the degree of financialization between countries. The 41-year sample period covers several major shocks to the oil market, including the 2008 financial crisis, the shale oil revolution, the 2014 OPEC "Price War," and the COVID-19 pandemic. Figure 7 presents the Spearman correlation matrix between the futures market variable and the FD and FI indices for the 12 countries. Due to the lack of high-frequency data, the correlations are calculated using annual data, and the background information and descriptive statistics of the two datasets are presented in Table 3 and Table 4, respectively. The results show a significant positive correlation over the 41-year period. The figure shows that it is appropriate to analyze the impact of financialization from the perspective of the corresponding futures markets for spot markets.

Meanwhile, supply-side advantages—such as the ability to influence global crude oil supply through production adjustments—may also have a significant impact on oil prices. Regarding the process of shaping global crude oil pricing hubs, no study has presented a comparative analysis of the relative importance of futures markets and production capacity. To address this gap, we examine the effects of the existence of crude oil futures markets and crude oil production levels on the temporal degree, temporal closeness, and temporal betweenness of crude oil spot markets. This approach draws on the framework of Kaufmann (2016), which investigates how country risk influences the speed of oil price adjustment.

Given that global oil production markets are highly susceptible to geopolitical

shocks (Saint Akadiri and Ozkan, 2025; Chiaramonte et al., 2025), we incorporate a geopolitical risk (GPR) index developed by Caldara and Iacoviello (2022), which is constructed based on news-based indicators, as a control variable. This allows us to test the robustness of our findings in the presence of geopolitical disturbances. Note that the GPR index from Caldara and Iacoviello (2022) does not include Kuwait, Oman, Qatar, and the UAE. Therefore, we set the GPR values for these four countries to 0 while applying a first-order difference to the GPR index for all other countries.

For additional control variables, we follow Zhu et al. (2021), who used the change rates of the Consumer Price Index (Δ CPI) and foreign exchange rates (Δ FX) in their study on the impacts of the trade pattern on risk spillover in oil markets. We also include official total reserves in the U.S. dollar (Δ In_Liq), published by the IMF, as a control variable. Owing to the "oil–dollar" link, the dollar reserves used for oil purchases constitute a significant item in the balance of payments of countries (Ivan et al., 2022). As such, dollar-based liquidity assets are relevant to our analysis.

Table 4 provides the background information on the panel and correlation data, and Table 5 presents the descriptive statistics for the panel correlation variables. Our regression models are as follows:

$$Degree_{it} = \alpha_{it} + \beta_1 Future_{it} + \beta_2 (Volume_{it}) + \gamma \mathbf{X}_{it} + \delta_t + \epsilon_{it}$$
(15)

$$Closeness_{it} = \alpha_{it} + \beta_1 Future_{it} + \beta_2 (Volume_{it}) + \gamma \mathbf{X}_{it} + \delta_t + \epsilon_{it}$$
(16)

Betweenness_{*it*} =
$$\alpha_{it} + \beta_1$$
Future_{*it*} + β_2 (Volume_{*it*}) + $\gamma \mathbf{X}_{it} + \delta_t + \epsilon_{it}$ (17)

where Degree_{it} is the normalized temporal degree of oil market *i* at time *t*. Closeness_{it} is the normalized temporal closeness of oil market *i* at time *t*. Betweenness_{it} is the normalized temporal betweenness of oil market *i* at time *t*. Volume_{it} is the volume of crude oil production in country *i* at time *t*. Future_{it} is a dummy variable that equals one if the spot oil market has a corresponding futures market and zero otherwise (Kaufmann, 2016). X_{it} is a vector of the control variables for country *i* at time *t*, ϵ_{it} signifies the stochastic disturbance term.

 α_{it} , β_1 , β_2 , β_3 , and γ are coefficients. δ_t is a time dummy.

For the dataset, we have *N* countries, with data ranging from January 2012 to February 2024, providing a comprehensive overview of the impact of these variables on the oil network.

In the regressions of the indicators, we apply fixed-month effects and clustered standard errors by month. This approach allows for more precise control and adjustment for month-specific effects and correlations in the data, eliminating confounding factors such as seasonal production variations and providing more reliable and accurate results.

Table 6 reports the regression results for the normalized temporal degree using a rolling window of 52 months. We consider fixed-month effects and clustered standard errors by month. In Model 1, the estimated coefficient for the presence of an oil futures market is 0.07, while the coefficient for oil production volume is 0.03. These results indicate that both the development of oil financial markets and increased oil production significantly contribute to the enhancement of the normalized temporal degree of crude oil spot markets within the global oil network. In other words, establishing highly liquid oil futures markets and maintaining substantial production volumes improve the information connectivity across spot markets.

To assess the robustness of this positive relationship and mitigate concerns regarding potential reverse causality, we lag key explanatory variables by one month and re-estimate the model. In Model 6, the lagged futures market variable continues to exhibit a coefficient of 0.07, and the coefficient for lagged production volume remains at 0.03—both nearly identical to the original estimates. Even after incorporating additional controls such as the rate of change in global liquidity and the GPR index, the positive relationship persists across all model specifications. This suggests that the observed effect is not merely driven by contemporaneous shocks, including geopolitical uncertainty, reinforcing the empirical validity of our initial hypothesis.

From a theoretical point of view, cross-border investors can transmit price signals from one market to another through capital flows. The price discovery function of oil futures markets allows spot markets to incorporate domestic demand conditions more rapidly in oil trade. In addition, futures markets act as new nodes in the global oil network, enriching the diversity and structure of the system and linking domestic spot markets to international markets. Consequently, financialization activities—particularly the introduction of oil derivatives—serve to improve the overall connectivity of crude oil spot markets. Our findings provide empirical support for the hypothesis that oil financialization enhances network integration.

We further confirm the positive impact of oil production on the temporal degree of spot markets. Under the supply–demand framework, crude oil, being a physical commodity with limited substitutes, is more susceptible to price control by producers. OPEC, as a representative cartel, illustrates the monopolistic role of suppliers in shaping global oil prices. For net oil exporters, expanding production strengthens their position in global supply and enables profit maximization through output adjustments. For net oil importers, increasing domestic production mitigates dependence on foreign oil and enhances national energy security. A prominent example is the U.S., where the shale oil revolution led to a sustained increase in domestic production, transforming the U.S. into a net oil exporter. This production growth expanded the reach of U.S. crude in global oil trade and elevated the global relevance of WTI crude oil prices. Hence, higher oil production implies greater market share and connectivity, allowing countries to establish more extensive links with other spot markets through trade volume control.

Table 7 presents the regression results for normalized temporal closeness, controlling for fixed-month effects and clustering standard errors at the monthly level. In Model 1, the coefficient for the existence of an oil futures market is 0.07. The lagged specification in Model 6 produces a nearly identical coefficient, reinforcing the robustness of the result. These findings suggest that the presence of oil futures markets significantly increases temporal closeness, indicating faster and more efficient information transmission within the global crude oil spot market network. In practice, certain spot markets report only one or a limited number of shipment prices per trading day. By contrast, futures markets rely on electronic trading platforms that generate continuous streams of price and volume data throughout the trading day, offering a more consistent and granular reflection of the country's oil market conditions. Through the use of highly liquid capital and highfrequency trading, such information is swiftly transmitted across markets and incorporated into domestic spot prices, thereby facilitating near-instantaneous market adjustments. In this context, the speed of adjustment in oil markets can be interpreted as a proxy for market efficiency. Markets that fail to respond promptly to publicly available information create arbitrage opportunities for speculative trading, which runs counter to the risk-management objectives of both economic policy and energy security (Vansteenkiste, 2011).

Unlike the results for normalized temporal degree, the relationship between oil production and normalized temporal closeness is not statistically significant across all model specifications. This suggests that variations in production levels do not necessarily accelerate the rate at which information is transmitted in spot markets. While we do not dispute OPEC's capacity to influence global oil prices through supply adjustments, the principal objective of pricing for OPEC and other major producers is to maximize national economic returns rather than to provide price signals or hedging instruments for other market participants. Consequently, pricing intentions in key exporting countries, particularly in the Middle East, are largely insulated from information originating in external markets.

This finding is consistent with the operational realities of oil markets, where production levels are relatively rigid in the short to medium term. On the one hand, logistical and storage constraints limit producers' ability to enact substantial output adjustments. On the other hand, price movements generally occur at a much higher frequency than production changes, leading to a mismatch in adjustment intervals between the two variables. Thus, increases in production alone do not directly contribute to faster information flow in spot markets. This interpretation is consistently supported by all model results reported in Table 7.

Table 8 presents the regression results for normalized betweenness centrality. In Model 1, the regression coefficient for the futures market variable is 0.056, while the coefficient for the production variable is 0.061. These results suggest that the establishment of futures markets for spot trading and the increase in oil production both contribute to enhancing a spot market's betweenness centrality within the global oil network. This positive relationship holds consistently across both the baseline regressions and the lagged specifications (Models 6 through 10).

After introducing control variables such as the change in liquidity and the change in the GPR index, the core results remain robust, indicating that the inclusion of these controls does not alter the main hypothesis. The stability of these findings underscores the reliability of the positive association. Betweenness captures the extent to which other nodes depend on a particular node to transmit information or resources. On the one hand, nodes with high betweenness act as critical conduits for markets with asymmetric information, effectively serving as "information bridges" Liu and Gong (2020). On the other hand, betweenness centrality also highlights the structural importance of a node in network security. Removing a key intermediary node can significantly reduce the efficiency of information flow across the entire network.

Nodes with high betweenness are frequently used and relied upon by other markets, becoming the most "busy" nodes within the global oil market network. Therefore, enhancing betweenness through the development of futures markets and increased oil production can increase the importance of a country's spot market in global information exchanges, thus exerting greater influence on international oil pricing.

7 Conclusion

This study systematically analyzes the impact of oil production and financialization, represented by oil futures markets, on the pricing power of crude oil spot markets. We particularly focus on their differential effects on 12 key oil spot markets in the global oil market network. Using weekly settlement prices of crude oil spot markets as data samples, we calculate the transfer entropy between the price returns to measure bidirectional causal relationships. Transfer entropy serves as a proxy for fundamental pricing power. A series of dynamic networks are constructed based on the direction and intensity of transfer entropy, and quantitative analysis is performed using network centrality indicators, including degree, closeness, and betweenness centrality. This approach reveals causal relationships between price information across oil markets, providing a new perspective on the identification of market pricing power. The results of the full sample demonstrate that the Brent and WTI markets occupy key positions in the global oil network.

Moreover, the rolling sample analysis illustrates the dynamic evolution of market centrality indicators (degree, closeness, and betweenness) over a span of 146 months. The results show that after the 2014 "Price War," the pricing power of the global oil market experienced significant turbulence. This suggests that shocks and restructurings from important events may disrupt the spillover of information in the global oil network, highlighting the need for regulators to pay special attention during such periods. The impacts of financialization and oil production on network centrality reveal that, although the establishment of oil futures markets significantly enhances market connectivity, transmission efficiency, and information mediation, increases in oil production do not significantly accelerate the speed of information spillovers. This positive relationship remains unaffected by geopolitical uncertainties and changes in official liquidity reserves. Thus, contrary to the conventional understanding of pricing power (Tudor and Anghel, 2021), our data suggest that futures markets exert a more comprehensive influence on the spillover capacity of information than oil production. The differences in influence channels help explain why Brent and WTI have remained stable centers over the long term, whereas Middle Eastern markets have not: financialization affects market behavior more rapidly and effectively than production levels.

To help enhance risk management capabilities and pricing power in oil spot markets, our findings support three policy recommendations for financial institutions and regulatory bodies involved in international oil trading and investment. First, we recommend strengthening the construction and regulation of oil financial derivative markets. Governments should increase support for oil futures and options markets, ensuring that these markets have adequate liquidity and security. This will enhance market transparency and information integration, providing more comprehensive and accurate forward-looking information to spot markets. Second, countries should establish stable oil production plans to maintain their influence within the global oil network. Specifically, governments should closely monitor fluctuations in national oil production, evaluate their position in the global oil supply chain, and implement emergency measures to prevent spot market failure in the event of drastic production changes, safeguarding investor interests. Finally, investment and regulatory bodies should enhance their monitoring of the information transmission capacity of key oil markets. Governments and market participants must pay close attention to the spillover capabilities of major oil markets. When these markets lose their reference value, continued reliance on their price signals should be avoided to protect the interests of the national oil market. In particular, during extreme economic conditions, oil market participants and economic information users should select price benchmarks from markets with strong pricing power and efficient information transmission.

Despite the progress made in this study in revealing the impact of oil financial derivative markets and oil production on oil pricing power, several limitations remain. First, the sample period is limited to 2010–2024. Future research could extend the time span to further explore the evolution of oil market pricing power over a longer time horizon. Second, although this study has examined the roles of futures markets and production, future research could incorporate additional financialization indicators to enrich the intrinsic definition of oil financialization and provide more micro-level insight into its impact on oil pricing power.

References

- Almutairi, Hossa, Pierru, Axel, and Smith, James L. Pandemic, ukraine, opec+ and strategic stockpiles: Taming the oil market in turbulent times. *Energy Economics*, 144:108319, 2025.
- Bachmeier, Lance J and Griffin, James M. Testing for market integration crude oil, coal, and natural gas. *The Energy Journal*, 27(2):55–71, 2006.
- Behar, Alberto and Ritz, Robert A. Opec vs us shale: Analyzing the shift to a market-share strategy. *Energy Economics*, 63:185–198, 2017.
- Behrendt, Simon, Dimpfl, Thomas, Peter, Franziska J, and Zimmermann, David J. Rtransferentropy—quantifying information flow between different time series using effective transfer entropy. *SoftwareX*, 10:100265, 2019.
- Bentzen, Jan. Does opec influence crude oil prices? testing for co-movements and causality between regional crude oil prices. *Applied Economics*, 39(11):1375–1385, 2007.
- Caldara, Dario and Iacoviello, Matteo. Measuring geopolitical risk. *American* economic review, 112(4):1194–1225, 2022.
- Caporale, Guglielmo Maria, Ciferri, Davide, and Girardi, Alessandro. Timevarying spot and futures oil price dynamics. *Scottish Journal of Political Economy*, 61(1):78–97, 2014.
- Chatziantoniou, Ioannis, Gabauer, David, and Gupta, Rangan. Integration and risk transmission in the market for crude oil: New evidence from a timevarying parameter frequency connectedness approach. *Resources Policy*, 84: 103729, 2023.
- Chiaramonte, Laura, Mecchia, Federico, Paltrinieri, Andrea, and Sclip, Alex. Geopolitical risk and energy markets: Past, present, and future. *Journal of Economic Surveys*, 2025.
- Choi, Insu and Kim, Woo Chang. Enhancing exchange-traded fund price predictions: Insights from information-theoretic networks and node embeddings. *Entropy*, 26(1):70, 2024.
- Cui, Jinxin and Maghyereh, Aktham. Higher-order moment risk connectedness and optimal investment strategies between international oil and commodity futures markets: Insights from the covid-19 pandemic and russia-ukraine conflict. *International Review of Financial Analysis*, 86:102520, 2023.
- De Graaff, Nana. The rise of non-western national oil companies: Transformation of the neoliberal global energy order? *Neoliberalism in Crisis*, pages 161–178, 2012.
- Diebold, Francis X and Yilmaz, Kamil. Measuring financial asset return and volatility spillovers, with application to global equity markets. *The Economic Journal*, 119(534):158–171, 2009.

- Elder, John, Miao, Hong, and Ramchander, Sanjay. Price discovery in crude oil futures. *Energy Economics*, 46:S18–S27, 2014.
- Engle, Robert F and Granger, Clive WJ. Co-integration and error correction: representation, estimation, and testing. *Econometrica*, pages 251–276, 1987.
- Fama, Eugene F. Efficient capital markets. Journal of Finance, 25(2):383-417, 1970.
- Fan, Ying, Zhang, Yue-Jun, Tsai, Hsien-Tang, and Wei, Yi-Ming. Estimating 'value at risk' of crude oil price and its spillover effect using the ged-garch approach. *Energy Economics*, 30(6):3156–3171, 2008.
- Feng, Yanhong, Wang, Xiaolei, Chen, Shuanglian, and Liu, Yanqiong. Impact of oil financialization on oil price fluctuation: A perspective of heterogeneity. *Energies*, 15(12):4294, 2022.
- Foster, Andrew J. Price discovery in oil markets: a time varying analysis of the 1990–1991 gulf conflict. *Energy Economics*, 18(3):231–246, 1996.
- Frino, Alex, Ibikunle, Gbenga, Mollica, Vito, and Steffen, Tom. The impact of commodity benchmarks on derivatives markets: The case of the dated brent assessment and brent futures. *Journal of Banking & Finance*, 95:27–43, 2018.
- Garbade, Kenneth D and Silber, William L. Price movements and price discovery in futures and cash markets. *The Review of Economics and Statistics*, pages 289–297, 1983.
- Hirshleifer, David. Residual risk, trading costs, and commodity futures risk premia. *The Review of Financial Studies*, 1(2):173–193, 1988.
- Hirshleifer, David. Futures trading, storage, and the division of risk: A multiperiod analysis. *The Economic Journal*, 99(397):700–719, 1989.
- Hirshleifer, David. Hedging pressure and futures price movements in a general equilibrium model. *Econometrica*, pages 411–428, 1990.
- Ivan, Miruna-Daniela, Banti, Chiara, and Kellard, Neil. Prime money market funds regulation, global liquidity, and the crude oil market. *Journal of International Money and Finance*, 127:102671, 2022.
- Ji, Qiang and Fan, Ying. Dynamic integration of world oil prices: A reinvestigation of globalisation vs. regionalisation. *Applied Energy*, 155:171–180, 2015.
- Ji, Qiang and Fan, Ying. Evolution of the world crude oil market integration: A graph theory analysis. *Energy Economics*, 53:90–100, 2016.
- Ji, Qiang and Zhang, Dayong. China's crude oil futures: Introduction and some stylized facts. *Finance Research Letters*, 28:376–380, 2019.
- Kaufmann, Robert K. Price differences among crude oils: the private costs of supply disruptions. *Energy Economics*, 56:1–8, 2016.

- Kaufmann, Robert K and Ullman, Ben. Oil prices, speculation, and fundamentals: Interpreting causal relations among spot and futures prices. *Energy Economics*, 31(4):550–558, 2009.
- Kaufmann, Robert K, Dees, Stephane, Karadeloglou, Pavlos, and Sanchez, Marcelo. Does opec matter? an econometric analysis of oil prices. *The Energy Journal*, 25(4):67–90, 2004.
- Kilian, Lutz. Not all oil price shocks are alike: Disentangling demand and supply shocks in the crude oil market. *American economic review*, 99(3):1053–1069, 2009.
- Kim, Hyoungshick and Anderson, Ross. Temporal node centrality in complex networks. *Physical Review E*, 85(2):026107, 2012.
- Lee, Chien-Chiang, Zhou, Hegang, Xu, Chao, and Zhang, Xiaoming. Dynamic spillover effects among international crude oil markets from the timefrequency perspective. *Resources Policy*, 80:103218, 2023.
- Liu, Tangyong and Gong, Xu. Analyzing time-varying volatility spillovers between the crude oil markets using a new method. *Energy Economics*, 87:104711, 2020.
- Ma, Richie Ruchuan, Xiong, Tao, and Bao, Yukun. The russia-saudi arabia oil price war during the covid-19 pandemic. *Energy Economics*, 102:105517, 2021.
- Monge, Manuel, Gil-Alana, Luis A, and Gracia, de Fernando Pérez. Us shale oil production and wti prices behaviour. *Energy*, 141:12–19, 2017.
- Niu, Hongli and Hu, Ziang. Information transmission and entropy-based network between chinese stock market and commodity futures market. *Resources Policy*, 74:102294, 2021.
- Quint, D. and Venditti, F. The influence of opec+ on oil prices: A quantitative assessment. *The Energy Journal*, 44(5):173–186, 2023. doi: 10.5547/01956574.44. 4.dqui. Original work published 2023.
- Razek, Noha HA and Michieka, Nyakundi M. Opec and non-opec production, global demand, and the financialization of oil. *Research in International Business and Finance*, 50:201–225, 2019.
- Saint Akadiri, Seyi and Ozkan, Oktay. Risk across the spectrum: Unpacking the nexus of global oil uncertainty, geopolitical tensions, energy volatility, and uschina trade tensions. *Energy Policy*, 202:114609, 2025.
- Scheitrum, Daniel P, Carter, Colin A, and Revoredo-Giha, Cesar. Wti and brent futures pricing structure. *Energy Economics*, 72:462–469, 2018.
- Schreiber, Thomas. Measuring information transfer. *Physical review letters*, 85(2): 461, 2000.

- Schwarz, Thomas V and Szakmary, Andrew C. Price discovery in petroleum markets: Arbitrage, cointegration, and the time interval of analysis. *The Journal* of Futures Markets (1986-1998), 14(2):147, 1994.
- Silverio, Renan and Szklo, Alexandre. The effect of the financial sector on the evolution of oil prices: Analysis of the contribution of the futures market to the price discovery process in the wti spot market. *Energy Economics*, 34(6): 1799–1808, 2012.
- Strange, Susan. States and Markets: An Introduction to International Political Economy. Pinter, London, 1988.
- Tudor, Cristiana and Anghel, Andrei. The financialization of crude oil markets and its impact on market efficiency: Evidence from the predictive ability and performance of technical trading strategies. *Energies*, 14(15):4485, 2021.
- Vansteenkiste, Isabel. What is driving oil futures prices? fundamentals versus speculation. *Fundamentals versus speculation*, 2011.
- Wu, Fei, Xiao, Xuanqi, Zhou, Xinyu, Zhang, Dayong, and Ji, Qiang. Complex risk contagions among large international energy firms: A multi-layer network analysis. *Energy Economics*, 114:106271, 2022.
- Zhang, Dayong, Ji, Qiang, and Kutan, Ali M. Dynamic transmission mechanisms in global crude oil prices: Estimation and implications. *Energy*, 175:1181–1193, 2019.
- Zhang, Hai-Ying, Ji, Qiang, and Fan, Ying. Competition, transmission and pattern evolution: A network analysis of global oil trade. *Energy Policy*, 73:312–322, 2014.
- Zhang, Qi, Hu, Yi, Jiao, Jianbin, and Wang, Shouyang. The impact of russia– ukraine war on crude oil prices: An emc framework. *Humanities and Social Sciences Communications*, 11(1):1–12, 2024.
- Zhou, Yang, Xie, Chi, Wang, Gang-Jin, Gong, Jue, Li, Zhao-Chen, and Zhu, You. Who dominate the information flowing between innovative and traditional financial assets? a multiscale entropy-based approach. *International Review of Economics & Finance*, 93:329–358, 2024.
- Zhou, Yuqin, Wu, Shan, and Zhang, Zeyi. Multidimensional risk spillovers among carbon, energy and nonferrous metals markets: Evidence from the quantile var network. *Energy Economics*, 114:106319, 2022.
- Zhu, Bo, Liu, Jiahao, Lin, Renda, and Chevallier, Julien. Cross-border systemic risk spillovers in the global oil system: Does the oil trade pattern matter? *Energy Economics*, 101:105395, 2021.
- Zhu, Bo, Deng, Yuanyue, and Hu, Xin. Global energy security: do internal and external risk spillovers matter? a multilayer network method. *Energy Economics*, 126:106961, 2023.

Table

	Country	Full Name	Unit	Future
Brent	UK	Brent Crude Oil Spot	USD/barrel	1
WTI	USA	WTI Crude Oil Spot	USD/barrel	1
Minas	Indonesia	Asia Minas Crude FOB Indonesia Cargo Spot	USD/barrel	0
Russia	Russia	Crude Oil ESPO Blend FOB	USD/barrel	0
Mexico	Mexico	Pemex Mexican Crude Oil Basket Price	USD/barrel	0
China	China	Daqing Crude Spot	USD/barrel	0
Oman	Oman	Oman Crude Spot	USD/barrel	1
Qatar	Qatar	Qatar Cargo Spot	USD/barrel	0
Kuwait	Kuwait	Kuwait Cargo Spot	USD/barrel	0
Tapis	Malaysia	Asia Tapis Crude FOB Kerteh Cargo Spot	USD/barrel	0
Dubai	UAE	Dubai Crude Spot	USD/barrel	1
Saudi Arabia	Saudi Arabia	Arab Light Crude Oil Spot	USD/barrel	0

Table 1: Background Information of Data

Notes: The table lists the spot prices for various crude oil benchmarks from different countries and regions.

Window=52												
	Br	Me	WT	Ru	Du	Ku	Qa	Om	Sa	Mi	Та	Cn
Degree	0.785	0.859	0.798	1.000	0.143	0.065	0.143	0.222	0.000	0.337	0.133	0.208
Closeness	0.781	0.609	1.000	0.843	0.017	0.066	0.194	0.252	0.107	0.560	0.000	0.065
Betweenness	1.000	0.766	0.840	0.616	0.000	0.135	0.372	0.116	0.305	0.568	0.192	0.244
Window=56												
	Br	Me	WT	Ru	Du	Ku	Qa	Om	Sa	Mi	Та	Cn
Degree	0.818	1.000	0.430	0.310	0.028	0.328	0.018	0.256	0.000	0.744	0.145	0.355
Closeness	1.000	0.487	0.245	0.376	0.319	0.338	0.114	0.623	0.132	0.669	0.129	0.000
Betweenness	1.000	0.571	0.301	0.267	0.000	0.238	0.012	0.162	0.036	0.371	0.059	0.257
Window=60												
	Br	Me	WT	Ru	Du	Ku	Qa	Om	Sa	Mi	Та	Cn
Degree	0.667	1.000	0.430	0.387	0.095	0.326	0.071	0.319	0.000	0.754	0.185	0.358
Closeness	1.000	0.713	0.403	0.615	0.565	0.447	0.306	0.873	0.256	0.915	0.257	0.000
Betweenness	1.000	0.783	0.360	0.483	0.080	0.549	0.000	0.451	0.010	0.549	0.228	0.350

 Table 2: Normalized Temporal Degree, Closeness and Betweenness (Full Samples)

Notes: Br - Brent, Me - Mexico, WT - WTI, Ru - Russia, Du - Dubai, Ku - Kuwait, Qa - Qatar, Om - Oman, Sa - Saudi Arabia, Mi - Minas, Ta - Tapis, Cn - China.

Variable	Label	Definition	Measurement	Source
Degree	Normalized Degree	Rolling calculated degree of each oil spot market	None	Author
Closeness	Normalized Closeness	Rolling calculated closeness of each oil spot market	None	Author
Betweenness	Normalized Betweenness	Rolling calculated betweenness of each oil spot market	None	Author
Future	Future Market	Dummy variable for oil futures market	None	EIA
Volume	Oil Production Volume	Monthly crude oil production (incl. conden-	kb/d (thousand	EIA
		sate), scaled down by a factor of 10000	barrels per day)	
Δ GPR	Δ GPR	Change of Geopolitical risk (GPR) index	None	Caldara and Ia- coviello (2022)
$\Delta \text{ CPI}$	Δ CPI	Change of Consumer Price Index	None	IMF
Δ FX	Δ FX	Change in exchange rates (domestic cur- rency per Euro)	None	IMF
$\Delta \ln_Liq$	Δ Ln(returns of Liquidity)	Log returns of Official total reserves	None	IMF
FD	Financial Development Index	Relative ranking of financial institutions and markets development across countries	None	IMF
FI	Financial Institutions Index	Relative ranking of financial institutions across countries	None	IMF

Table 3: Panel and Correlation Data

Notes: EIA is the United States Energy Information Administration. IMF is the International Monetary Fund.

Variable	Mean	Std. Dev	Min	Q1	Median	Q3	Max
Normalized Degree	0.372	0.322	0.000	0.091	0.293	0.579	1.000
Normalized Closeness	0.490	0.318	0.000	0.226	0.477	0.758	1.000
Normalized Betweenness	0.326	0.333	0.000	0.048	0.197	0.535	1.000
Future	0.333	0.472	0.000	0.000	0.000	1.000	1.000
Volume	0.390	0.377	0.045	0.095	0.255	0.529	1.330
Δln_Liq	0.003	0.037	-0.394	-0.009	0.002	0.015	0.340
ΔGPR	0.002	0.294	-3.972	-0.022	0.000	0.021	3.405
ΔCPI	2.816	2.595	-3.982	1.364	2.495	3.742	17.800
ΔFX	0.000	0.025	-0.232	-0.011	0.001	0.012	0.357
FD	0.442	0.203	0.000	0.302	0.401	0.540	0.956
FI	0.448	0.213	0.000	0.310	0.393	0.563	0.943

Table 4: Descriptive Statistics of the Panel and Correlation Data

Notes: The table provides descriptive statistics for the panel data, including mean, standard deviation, minimum, first quartile (Q1), median, third quartile (Q3), and maximum values for each variable.

	Obs.	Mean	Std. Dev.	Skewness	Kurtosis-3	J-B	ARCH(20)	ADF
Brent	733	0.0002	0.05	-1.08	15.24	7238.60	264.23	-9.68***
WTI	733	0.0001	0.10	-2.91	226.52	1568134.31	258.95	-10.16***
Minas	733	0.0001	0.06	-0.64	16.95	8829.30	259.10	-8.16***
Russia	733	0.0002	0.05	-0.68	8.76	2399.79	292.89	-9.50***
Mexico	733	0.0002	0.07	0.03	20.34	12629.69	333.61	-9.79***
China	733	0.0002	0.05	-1.25	11.24	4049.78	211.59	-9.12***
Oman	733	0.0002	0.05	-0.67	6.84	1483.71	141.42	-9.46***
Qatar	733	0.0002	0.05	-0.41	7.75	1741.87	158.40	-8.94***
Kuwait	733	0.0002	0.05	-0.54	7.55	1869.82	189.34	-8.99***
Tapis	733	0.0002	0.05	-0.62	5.52	977.27	136.41	-8.91***
Dubai	733	0.0002	0.05	-0.56	6.37	1276.65	122.27	-9.41***
Saudi Arabia	733	0.0002	0.05	-0.48	4.44	631.10	151.29	-8.77***

 Table 5: Descriptive Results

Notes: ***, **, * denote significance at 1%, 5%, and 10% significance levels, respectively.

Variables		Norn	nalized D	egree		lag=1		Normalized Degree			
	(1)	(2)	(3)	(4)	(5)		(6)	(7)	(8)	(9)	(10)
Future	0.071***	0.070***	0.070***	0.082***	0.083***	Future_lag	0.069***	0.069***	0.069***	0.082***	0.082***
	(6.79)	(6.62)	(6.62)	(6.71)	(6.72)		(6.67)	(6.51)	(6.53)	(6.60)	(6.59)
Volume	0.031^{*}	0.033**	0.033**	0.018	0.018	Volume_lag	0.030^{*}	0.032^{*}	0.032^{*}	0.017	0.017
	(1.89)	(2.01)	(2.01)	(1.16)	(1.15)		(1.83)	(1.94)	(1.95)	(1.05)	(1.05)
∆ln_Liq		0.409	0.408	0.423	0.425	Δln_Liq_lag		0.398	0.396	0.412	0.412
		(1.50)	(1.50)	(1.56)	(1.57)			(1.47)	(1.47)	(1.53)	(1.53)
ΔGPR			-0.010*	-0.009*	-0.012	ΔGPR_{lag}			-0.017	-0.016	-0.017
			(-0.39)	(-0.34)	(-0.44)				(-0.63)	(-0.57)	(-0.58)
ΔCPI				0.010^{***}	0.010^{***}	∆CPI_lag				0.011^{***}	0.011^{***}
				(3.63)	(3.66)					(3.76)	(3.76)
ΔFX					-0.165	ΔFX_{lag}					-0.062
					(-0.45)						(-0.16)
Fixed Effects	Yes	Yes	Yes	Yes	Yes	Fixed Effects	Yes	Yes	Yes	Yes	Yes
Ν	1,752	1,752	1,752	1,752	1,752	Ν	1,740	1,740	1,740	1,740	1,740
R ²	0.110	0.112	0.112	0.116	0.117	R ²	0.107	0.109	0.109	0.115	0.115

 Table 6: Regression of Normalized Degree

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. T-statistics are in parentheses. $\Delta \log$ _Liq represents the change in the logarithm of liquidity returns.

Standard errors are clustered at the country level. All models include time fixed effects.

Variables		Norma	alized Clo	oseness		lag=1		Normalized Closeness			
	(1)	(2)	(3)	(4)	(5)		(6)	(7)	(8)	(9)	(10)
Future	0.069***	0.069***	0.069***	0.098***	0.098***	Future_lag	0.067***	0.067***	0.067***	0.097***	0.097***
	(5.79)	(5.71)	(5.70)	(6.99)	(7.02)		(5.67)	(5.59)	(5.60)	(7.05)	(7.08)
Volume	0.026	0.027	0.027	-0.007	-0.007	Volume_lag	0.026	0.027	0.027	-0.009	-0.009
	(1.41)	(1.46)	(1.46)	(-0.36)	(-0.36)		(1.40)	(1.45)	(1.45)	(-0.44)	(-0.44)
∆ln_Liq		0.220	0.220	0.255	0.254	∆ln_Liq_lag		0.206	0.205	0.242	0.239
		(1.02)	(1.02)	(1.27)	(1.26)			(1.00)	(0.99)	(1.25)	(1.23)
ΔGPR			0.000	0.003	0.006	ΔGPR_{lag}			-0.012	-0.008	-0.003
			(0.01)	(0.13)	(-0.22)				(-0.43)	(-0.31)	(-0.10)
ΔCPI				0.025***	0.025***	∆CPI_lag				0.026***	0.026***
				(7.69)	(7.65)					(8.21)	(8.09)
ΔFX					0.134	ΔFX_{lag}					0.260
					(0.38)						(0.65)
Fixed Effects	Yes	Yes	Yes	Yes	Yes	Fixed Effects	Yes	Yes	Yes	Yes	Yes
Ν	1,752	1,752	1,752	1,752	1,752	Ν	1,740	1,740	1,740	1,740	1,740
R ²	0.129	0.129	0.129	0.158	0.159	R ²	0.128	0.129	0.129	0.161	0.162

 Table 7: Regression of Normalized Closeness

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. T-statistics in parentheses. $\Delta \log$ _Liq represents the change in the logarithm of liquidity returns. Standard errors are clustered at the country level. All models include time fixed effects.

Variables		Normal	ized Betw	veenness		lag=1	Normalized Betweenness				
	(1)	(2)	(3)	(4)	(5)		(6)	(7)	(8)	(9)	(10)
Future	0.056***	0.056***	0.056***	0.064***	0.065***	Future_lag	0.054***	0.053***	0.053***	0.063***	0.064***
	(3.84)	(3.78)	(3.78)	(3.89)	(3.97)		(3.70)	(3.60)	(3.60)	(3.86)	(3.92)
Volume	0.061***	0.062***	0.063***	0.053***	0.052***	Volume_lag	0.060***	0.062***	0.062***	0.049^{***}	0.049***
	(3.64)	(3.72)	(3.73)	(3.05)	(3.06)		(3.56)	(3.68)	(3.68)	(2.86)	(2.86)
∆ln_Liq		0.216	0.215	0.225	0.233	∆ln_Liq_lag		0.359	0.359	0.372	0.377
		(0.94)	(0.94)	(0.98)	(1.01)			(1.53)	(1.53)	(1.58)	(1.60)
ΔGPR			-0.012	-0.012	-0.025	ΔGPR_{lag}			-0.000	0.001	-0.009
			(-0.35)	(-0.35)	(-0.66)				(-0.01)	(0.03)	(-0.23)
ΔCPI				0.007^{**}	0.007^{**}	∆CPI_lag				0.009^{**}	0.009**
				(2.07)	(2.06)					(2.59)	(2.59)
ΔFX					-0.668	ΔFX_{lag}					-0.468
					(-1.58)						(-1.08)
Fixed Effects	Yes	Yes	Yes	Yes	Yes	Fixed Effects	Yes	Yes	Yes	Yes	Yes
Ν	1,752	1,752	1,752	1,752	1,752	Ν	1,740	1,740	1,740	1,740	1,740
R ²	0.090	0.090	0.090	0.092	0.094	R ²	0.088	0.089	0.089	0.093	0.093

 Table 8: Regression of Normalized Betweenness

Notes: *** p<0.01, ** p<0.05, * p<0.1. T-statistics are in parentheses. Standard errors are clustered at the country level. All models include time fixed effects.

Figure



Figure_1. Oil Prices



Figure_2. Returns of the Oil Prices



Figure_3. Network Slices



Figure_6. Temporal Betweeness



Figure_7. Correlation between the Futures Variable and the FD and FI Index